Author's response to reviews

Title: Comparative Effectiveness Using Two Different Interbody Fusion Methods for Transforaminal Lumbar Interbody Fusion: Cages Versus Merely Morselized Impacted Bone Graft

Authors:

Chaoliang Lv (lvchaolianggk@163.com)
Xianzhou Li (fangyuegk3@163.com)
Haicheng Zhang (tuchongqi6666@163.com)
Junrong Lv (duanhonggk@163.com)
Hongmei Zhang (442326621@qq.com)

Version: 2
Date: 16 March 2015

Author's response to reviews:

According to editorial requests, we check the submission again and make sure it comply with editorial policies. We add the list of abbreviations, revise our manuscript to include line and page numbers.